Author's response to reviews

Title: Iatrogenic left ventricular-right atrial communication after tricuspid annuloplasty; A case report

Authors:

Eiki Tayama (e.tayama@kyumed.jp)
Yukihiro Tomita (ytomita@kyumed.jp)
Ken-ichi Imasaka (imasaka@kyumed.jp)
Takanori Kono (tesshin@kyumed.jp)

Version: 2 Date: 3 June 2014

Author's response to reviews: see over
June 3, 2014

Dr. Vipin Zamvar, Royal Infirmary of Edinburgh
Dr. David Taggart, John Radcliffe Hospital

Editors-in-Chief
Journal of Cardiothoracic Surgery

Dear Chief Editors;

Thank you very much for reviewing our manuscript entitled “Iatrogenic left ventricular-right atrial communication after tricuspid annuloplasty; A case report” for consideration for publication as a case report in Journal of Cardiothoracic Surgery.

As your suggestion, we made following changes. All changes were addressed with underlines in the revised manuscript.

1. Page 1~2
   I added all co-authors’ e-mail addresses.

   Co-authors’s e-mail addresses
   
   Y. Tomita: ytomita@kyumed.jp
   K. Imasaka: imasaka@kyumed.jp
   T. Kono: tesshin@kyumed.jp

2. Page 4 and 7 (in Abstract and Case Presentation)
   I added following sentence regarding to Ethnicity. Although the sentences were added in both abstract an case presentation, “Consent” section was left as it was (page 8-9).

   Written informed consent was obtained from the patient for publication of this Case Report and any accompanying images.
Again, we declare that this manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and approved this submission. The authors report no conflicts of interest.

We look forward to hearing from you at your earliest convenience.

Sincerely,

Eiki Tayama M.D., Ph.D.
Department of Cardiovascular Surgery
Clinical Research Institute
National Hospital Organization Kyushu Medical Center
1-8-1 Jigyohama, Chuo-ku
Fukuoka City 810-8563, Japan
Tel: +81-92-852-0700
Fax +81-92-847-8802
E-mail: e.tayama@kyumed.jp